Author's response to reviews

Title: Microbicide excipients can greatly increase susceptibility to genital herpes transmission in the mouse

Authors:

Thomas R. Moench (tmoench@reprotect.com)
Russell J. Mumper (mumper@email.unc.edu)
Timothy E. Hoen (hoen@jhu.edu)
Mianmian Sun (mmsun@email.unc.edu)
Richard A. Cone (cone@jhu.edu)

Version: 5 Date: 16 November 2010

Author's response to reviews:

These were addressed earlier.